Author's response to reviews

Title: Transient Small Bowel Intussusception in an Adult: case report with intraoperative video and literature review

Authors:

Hager Aref (dr.hageraref@gmail.com)
Abrar Nawawi (ladybird_1403@hotmail.com)
Abdulmalik Altaf (altaf12345@yahoo.com)
Murad Aljiffry (dr.aljiffry@gmail.com)

Version: 6
Date: 16 February 2015

Author's response to reviews: see over
8-12-2014

Tom Rowles
Executive Editor of the BMC Surgery Journal

Dear Mr. Tom Rowles,

On behalf of all the authors, I would like to ask you to consider our manuscript entitled “Transient Small Bowel Intussusception in an Adult: case report with intraoperative video and literature review” for publication in the (BMC Surgery Journal) as a case report. This case report would emphasize on a condition that is rare and easily missed during the diagnosis process of cases with acute abdomen. The author highlights the importance of considering Transient Intussusception one of the differential diagnosis when a patient comes with acute abdomen and other more common conditions are rolled out. Interestingly, we have captured an intraoperative video of small bowel loop intussuscepting and reducing spontaneously, making this case report the first case report presented with an actual video tapping of the condition. The patient have provided permission to publish these features of his case, and the identity of the patient has been protected.

We feel that the findings described in this case report will be of special interest to the readers of the (BMC Surgery Journal).

This manuscript has not been published and is not under consideration for publication elsewhere. All the authors have read the manuscript and have approved this submission. The authors report no conflicts of interest.

Sincerely,

Corresponding author
Dr. Hager Aref,
Department of Surgery, Faculty of Medicine, King Abdulaziz University Hospital;
P.O. Box 1716, Jeddah 21441, Saudi Arabia
Telephone number (+966) (012) 6401000, Ext 18209